Author's response to reviews

Title: A Comparative Evaluation of the Effect of Internet-based CME Delivery Format on Satisfaction, Knowledge and Confidence

Authors:

Vernon R Curran (vcurran@mun.ca)
Lisa Fleet (lfleet@mun.ca)
Fran Kirby (fkirby@mun.ca)

Version: 3 Date: 21 December 2009

Author's response to reviews: see over
December 21, 2009

Nina Titmus
Scientific Editor
BMC-series Journals
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HL

Dear Ms. Titmus:

Please find attached a revised, final version of the manuscript titled “A Comparative Evaluation of the Effect of Internet-based CME Delivery Format on Satisfaction, Knowledge and Confidence”. The following revisions have been incorporated in this final version of the manuscript to address the reviewers’ comments:

- Total number of participants in each study group has been added at the beginning of the results section

- Additional comments added in the methods sections to indicate that knowledge and confidence assessments covered subject matter in the course.

- Footnote added to Table 3 for interpreting mean responses.

- Revised statement added on page 12 to define N.

If there are further queries, please contact me.

Thank you.

Sincerely,

Vernon Curran, PhD
Associate Professor